Author's response to reviews

Title: Time Trends of Baseline Demographics and Clinical Characteristics of HIV infected Children Enrolled in Care and Treatment Service in Dar es Salaam, Tanzania

Authors:

david M sando (dsando.tz@gmail.com)
Donna Spiegelman (stdls@hsph.harvard.edu)
Lameck Machumi (lmachumi@mdh-tz.org)
Mary Mwanyika-Sando (mmwanyika.sando@gmail.com)
Eric Aris (aeric@mdh-tz.org)
Aisa Muya (aisamuya@gmail.com)
Elizabeth Jackson (ejackson@hsph.harvard.edu)
Till Baernighausen (tbaernig@hsph.harvard.edu)
Ellen Hertzmark (stleh@channing.harvard.edu)
Guerino Chalamila (gchalamilla@gmail.com)
Wafaie Fawzi (mina@hsph.harvard.edu)

Version: 2 Date: 11 August 2014

Author's response to reviews:

I have now included emails of all authors in the title page
I have added competing interest after conclusion section
Funding statement is included in the acknowledgment section
Ethics statement is stated at the end of the method section
Authors contribution is included before the acknowledgment
Line numbers have also been added